Author’s response to reviews

Title: Treatment outcomes and antiretroviral uptake in Multidrug-Resistant Tuberculosis and HIV coinfected Patients in Sub Saharan Africa. A Systematic Review and Meta-Analysis

Authors:

Elvis Dzelamonyuy Chem (evlisc01@gmail.com)

Marie Van Hout (M.C.VanHout@ljmu.ac.uk)

Vivian Hope (V.D.Hope@ljmu.ac.uk)

Version: 2 Date: 24 Jul 2019

Author’s response to reviews:

Author’s response to Editor's comments

Title: “Treatment outcomes and antiretroviral uptake in Multidrug-Resistant Tuberculosis and HIV coinfected Patients in Sub Saharan Africa. A Systematic Review and Meta-Analysis “

Authors:

Elvis Dzelamonyuy Chem: (evlisc01@gmail.com)

Marie Claire Van Hout: (M.C.VanHout@ljmu.ac.uk)

Vivian Hope: (V.D.Hope@ljmu.ac.uk)

Version 2: Date: 24/07/2019

Author’s response to reviews:

Elvis Dzelamonyuy Chem
c/o Public Health Institute
Faculty of Health, Education and Community
Liverpool John Moores University
Subject: Revision and resubmission of manuscript INFD-D-19-00981

Dear Oliver,

Thank you for your response to our manuscript submission and your request for additional amendments. We have revised the paper to reflect your request. The amendments have been approved by all co-authors. We will appreciate a prompt reply from you at your earliest convenience.

Sincerely

Elvis Dzelamonyuy Chem